Author's response to reviews

Title: Acute abdomen and renal failure due to bilateral renal infarction in a woman: a case report

Authors:

khaireddine bouassida (khayri14@hotmail.com)
wissem hmida (khayri14@hotmail.com)
amira zairi (khayri14@hotmail.com)
adnen hidoussi (khayri14@hotmail.com)
faouzi mallat (khayri14@hotmail.com)
mehdi jaidane (khayri14@hotmail.com)
khaled hani (khayri14@hotmail.com)
faouzi ali mosbah (khayri14@hotmail.com)

Version: 5 Date: 16 February 2012

Author's response to reviews:

We would thank associate editor for their informative comments with regards to our article. Please find below responses

1. In figure 1, the arrow indicates a peri-renal fluid collection.

2. In figure 2, left panel, the arrow is removed and shows parenchymal perfusion defects of both kidneys related to segmental distal arterial lesions.
   The right hand panel, the arrow show also parenchymal perfusion defects of both kidneys related to segmental distal arterial lesions.

3. In figure 3, we changed the illustration by another which is clearer and 3 small arrows were added around the thrombus.

4. Figure 2 legend was modified.

5. We also include ethnicity of the patient both in the abstract and case presentation section.

6. - We replace the header, The Case by Case Presentation.

7. - We include one discussion section only.

8. - We move the figure and Table legend section after the reference list.

9. - We include the reference list at the end of the main manuscript document.